Author’s response to reviews

Title: Application of Protection Motivation Theory to Clinical Trial Enrolment for Pediatric Chronic Conditions

Authors:
Stephanie Brooks (skowal@ualberta.ca)
Tania Bubela (tbubela@sfu.ca)

Version: 2 Date: 27 Feb 2020

Author’s response to reviews:

-- Please ensure that all figures are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
ALL FIGURES AND TABLES ARE NOW CORRECTLY CITED IN TEXT

-- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
HAVE ADDED WRITTEN CONSENT

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
HAVE ADDED THAT FUNDERS HAD NO ROLE

-- Please ensure that additional files are carried over from the previous version of your manuscript.
HAVE CARRIED FORWARD TABLE AND FIGRES AND UPDATED THE MANUSCRIPT

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
ALL FILES ARE CLEAN VERSIONS